Author's response to reviews

Title: Prenatal Micronutrient Supplementation and Postpartum Depressive Symptoms in a Pregnancy Cohort

Authors:

Brenda MY Leung (bleun@ucalgary.ca)
Bonnie J Kaplan (kaplan@ucalgary.ca)
Catherine J Field (catherine.field@ualberta.ca)
Suzanne Tough (suzanne.tough@albertahealthservices.ca)
Misha Eliasziw (Misha.Eliasziw@tufts.edu)
Mariel F Gomez (fajergom@ualberta.ca)
Linda J McCargar (linda.mccargar@afhe.ualberta.ca)
Lisa Gagnon (lisa.gagnon@albertahealthservices.ca)

Version: 5 Date: 4 December 2012

Author's response to reviews:

To the Editors:

The suggested sentence change is much appreciated; the new wording is much more concise. It is shaded gray on pages 8-9.